Author's response to reviews

Title: Potential risk factors of diabetic neuropathy: a case control study

Authors:

Fargol Booya (emrc@tums.sina.ac.ir)
Bagher Larijani (emrc@tums.sina.ac.ir)
Mohammad Pajouhi (emrc@tums.sina.ac.ir)
Mahdi Nooraei (emrc@tums.sina.ac.ir)
Jamshid Lotfi (emrc@tums.sina.ac.ir)
Fatemeh Bandarian (emrc@tums.sina.ac.ir)

Version: 3 Date: 27 September 2005

Author's response to reviews: see over
Dear Editor
Thank you for your nice and faithful comments on the manuscript "Potential risk factors of diabetic neuropathy: a cross sectional study" (MS#1315091906736316). We tried to revise the manuscript according to the reviewer comment as more as possible, hope to satisfy them. Also a native English speaker copyedited the manuscript. The changes have been highlighted in the text and the revised version of the manuscript has been attached. The point by point response to the reviewers' questions is as follow:

**Reviewer#1: James Reed**

*Minor essential revisions:*
1- The first sentence was edited.
2- p-values were included in table 1.
3- The requested data was provided in table 2.
4- The values have been shown in table 3.

*Discretionary Revisions:*
Figure 2 was explained completely in the result section: page 6, line 18.

**Reviewer#2: Satoshi Kuwabara**

*Major compulsory revisions:*
The result of multivariate analysis for significant variables has been included in table 3.

*Minor essential revisions:*
1-EMG-NCV was corrected in the entire text and highlighted. The electrodiagnostic criteria for neuropathy were described in the text: page 5, line 4.

*Minor essential revisions:*
ACEI was spelled out in the abstract

**Reviewer#3: Peter J Dyck**

The result of multivariate analysis for significant variables has been included in table 3.
Abstract, introduction and discussion were edited.

Kind Regards,
Bagher Larijani, MD